# Rituximab versus tocilizumab and B-cell status in TNF-alpha inadequate-responder rheumatoid arthritis patients: the R4-RA RCT

Frances Humby,<sup>1</sup> Patrick Durez,<sup>2</sup> Maya H Buch,<sup>3</sup> Myles J Lewis,<sup>1</sup> Michele Bombardieri,<sup>1</sup> Christopher John,<sup>1</sup> Hasan Rizvi,<sup>4</sup> Louise Warren,<sup>1</sup> Joanna Peel,<sup>1</sup> Liliane Fossati-Jimack,<sup>1</sup> Rebecca E Hands,<sup>1</sup> Giovanni Giorli,<sup>1</sup> Felice Rivellese,<sup>1</sup> Juan D Cañete,<sup>5</sup> Peter C Taylor,<sup>6</sup> Peter Sasieni,<sup>7</sup> João E Fonseca,<sup>8</sup> Ernest Choy<sup>9</sup> and Costantino Pitzalis<sup>1\*</sup>

- <sup>1</sup>Centre for Experimental Medicine and Rheumatology, Queen Mary University of London, London, UK
- <sup>2</sup>Institute of Experimental and Clinical Research, Université Catholique de Louvain, Louvain, Belgium
- <sup>3</sup>Clinical Trials Research Unit, University of Leeds, Leeds, UK
- <sup>4</sup>Blizard Institute, Barts Health NHS Trust, London, UK
- <sup>5</sup>Arthritis Unit, L'Institut d'Investigacions Biomèdiques, Barcelona, Spain
- <sup>6</sup>Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK
- <sup>7</sup>King's Clinical Trial Unit, King's College London, London, UK
- <sup>8</sup>Instituto de Medicina Molecular, University of Lisbon, Lisbon, Portugal
- <sup>9</sup>Section of Rheumatology, Cardiff University School of Medicine, Cardiff University, Cardiff, UK

\*Corresponding author c.pitzalis@qmul.ac.uk

**Declared competing interests of authors:** Frances Humby reports grants and personal fees from Pfizer Inc. (New York, NY, USA), and personal fees and advisory fees from F. Hoffman La-Roche Ltd (Basel, Switzerland) outside the submitted work. Maya Buch reports grants and personal fees from Pfizer Inc., grants from F. Hoffman La-Roche Ltd, Chugai Pharmaceutical Company (Chuo City, Tokyo, Japan) and UCB (Brussels, Belgium), and personal fees from Pfizer Inc., AbbVie Inc. (North Chicago, IL, USA), Sandoz (Holkirchen, Germany), Sanofi (Paris, France), Merck Serono (Darmstadt, Germany) and Eli Lilly and Company (Indianapolis, IN, USA) outside the submitted work. Juan D Cañete reports personal fees from Eli Lilly and Company, Janssen Pharmaceutica (Beerse, Belgium), Mylan (Canonsburg, PA, USA), Novartis Pharmaceuticals (Basel, Switzerland) and UCB, and educational actions from Gilead (Foster City, CA, USA) outside the submitted work. Peter Taylor reports personal fees from F. Hoffman La-Roche Ltd, AbbVie Inc., Pfizer Inc. and UCB outside the submitted work. Peter Sasieni is director of King's Clinical Trials Unit (2017 to present), which receives core funding from the National Institute for Health and Care Research. João Fonseca reports grants from AbbVie Inc., Hospira (Lake Forest, IL, USA),

Janssen Pharmaceutica, Lilly, Merck, Sharp & Dohme Corp. (Kenilworth, NJ, USA), Novartis Pharmaceuticals, Pfizer Inc. and UCB, and personal fees and speaker fees from AbbVie Inc., Janssen Pharmaceutica, Lilly, Novartis Pharmaceuticals, Pfizer Inc. and F. Hoffman La-Roche Ltd outside the submitted work. João Fonseca also reports personal fees and advisory fees from Amgen Inc. (Thousand Oaks, CA, USA), Celltrion (Incheon, The Republic of Korea), outside the submitted work. Ernest Choy reports personal fees from AbbVie Inc., Amgen Inc., Bristol Myer Squibb™ (New York, NY, USA), Celgene Corp. (Summit, NJ, USA), Chugai Pharma Ltd, Eli Lilly and Company, Janssen Pharmaceutica, ObsEva (Chemin des Auix, Switzerland), Regeneron Pharmaceuticals (Tarrytown, NY, USA), Sanofi, SynAct Pharma (Lund, Swden) and Tonix Pharmaceuticals (Chatham, NJ, USA), and grants and personal fees from Bio-Cancer Treatment International (Hong Kong), NovImmune SA (Geneva, Switzerland), Pfizer Inc., F. Hoffman La-Roche Ltd and UCB Pharma Ltd., outside the submitted work. Costantino Pitzalis reports grants, personal fees and research support/scientific advisory board from AbbVie, Bristol Myers Squibb, Celgene Corp., Janssen/ Johnson & Johnson (New Brunswick, NJ, USA); personal fees and research support/scientific advisory board from AstraZeneca plc (Cambridge, UK)/MedImmune (Gaithersburg, MA, USA); grants, personal fees, non-financial support and research support/scientific advisory board from Pfizer Inc., F. Hoffman La-Roche Ltd, Genentech (South San Francisco, CA, USA) and Chugai Pharmaceutical Company; and personal fees and research support/scientific advisory board from AstraZeneca plc/MedImmune and UCB, outside the submitted work. In addition, the institution with which Costantino Pitzalis is affiliated, Queen Mary University of London (QMUL), has a patent from the original academic pilot study, on which the R4-RA (A Randomised, open-labelled study in anti-TNFalpha inadequate responders to investigate the mechanisms for Response, Resistance to Rituximab versus Tocilizumab in Rheumatoid Arthritis patients) trial was based, which identified B-cell status and associated expression profiles in the synovial biopsy as a good predictor of patient response to treatment with rituximab. The degree of association and predictive accuracy found in that study was high to the 94% level. These findings were intellectual property protected by QMUL (PCT number PCT/GB2015/052088; patent grant pending).

Published August 2022 DOI: 10.3310/GOPL1729

## **Plain English summary**

The R4-RA RCT Efficacy and Mechanism Evaluation 2022; Vol. 9: No. 7 DOI: 10.3310/GOPL1729

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

R heumatoid arthritis is a common disease in which the immune system attacks the joints, causing pain, stiffness and swelling.

In 40% of patients, conventional chemical drugs do not work and expensive 'biological' therapies are prescribed; however, these therapies can have serious side effects. The first biological drug to be prescribed is usually a type called a tumour necrosis factor inhibitor drug. If that fails, a second biological drug, rituximab (MabThera, F. Hoffman La-Roche Ltd, Basel, Switzerland), is recommended.

Rituximab reduces inflammation by attacking a specific type of immune cell (B cells). However, in about half of patients it has been found that, when the joint lining is examined under a microscope, B cells are present in only small numbers or are completely absent.

We reasoned that these patients, said to be B-cell poor, would respond better to another drug, tociluzimab, which works by a different mechanism. If this is the case, it might be possible, by examining a small piece of tissue from the lining of the joint, to predict which patients will respond better to each drug.

We examined the joint lining from 164 rheumatoid arthritis patients from hospitals in the UK, Belgium, Italy, Portugal and Spain who had not responded to tumour necrosis factor inhibitors. Of these patients, 79 were found to have few or no B cells and are described as B-cell poor. Approximately equal numbers of patients were treated with rituximab or tocilizumab (RoActemra, F. Hoffman La-Roche Ltd, Basel, Switzerland). After 16 weeks of treatment, we evaluated the number of tender and/or swollen joints.

We found that disease improvement in B-cell-poor patients was slightly better with tocilizumab than with rituximab, but that the difference was not enough to be sure that it did not arise by chance. However, when we assessed the joint lining tissue using newer, molecular methods and when we measured different outcomes, we found some promising results.

This means that we cannot yet be sure which drugs should be offered to which patients, but the results of this study could lead to further research to improve how we test and treat patients to get the best improvement in their arthritis.

<sup>©</sup> Queen's Printer and Controller of HMSO 2022. This work was produced by Humby *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health and Care Research, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

## **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb<sup>™</sup> (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme.

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME programme**

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 11/100/76. The contractual start date was in December 2012. The final report began editorial review in September 2019 and was accepted for publication in May 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2022. This work was produced by Humby *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health and Care Research, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk